

Citation: Kumar GS, Pezzi C, Wien S, Mamo B, Scott K, Payton C, et al. (2020) Health of Special Immigrant Visa holders from Iraq and Afghanistan after arrival into the United States using Domestic Medical Examination data, 2014–2016: A crosssectional analysis. PLoS Med 17(3): e1003083. https://doi.org/10.1371/journal.pmed.1003083

Academic Editor: Kolitha Wickramage, International Organization for Migration, SRI LANKA

Received: October 4, 2019

Accepted: February 24, 2020

Published: March 31, 2020

Copyright: This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative</u> <u>Commons CC0</u> public domain dedication.

Data Availability Statement: Because of ethical and legal considerations, we are unable to share de-identified individual-level data; only aggregate data are included in the manuscript. All relevant data are presented in the manuscript.

Funding: The state, county, and university coauthors (BM, KS, CP, KU, SH, LK, NC, JM, MT, JA, BK, and RF) on this manuscript received some funding from CDC-funded non-research **RESEARCH ARTICLE**

Health of Special Immigrant Visa holders from Iraq and Afghanistan after arrival into the United States using Domestic Medical Examination data, 2014–2016: A crosssectional analysis

Gayathri S. Kumaro¹*, Clelia Pezzi¹, Simone Wien¹, Blain Mamo², Kevin Scott³, Colleen Payton^{3#a}, Kailey Urban², Stephen Hughes⁴, Lori Kennedy⁵, Nuny Cabanting⁶, Jessica Montour^{7#b}, Melissa Titus⁸, Jenny Aguirre^{9#c}, Breanna Kawasaki⁵, Rebecca Ford^{10#d}, Emily S. Jentes¹

1 Centers for Disease Control and Prevention, Division of Global Migration and Quarantine, Immigrant, Refugee, and Migrant Health Branch, Atlanta, Georgia, United States of America, 2 Minnesota Department of Health, Saint Paul, Minnesota, United States of America, 3 Thomas Jefferson University, Philadelphia, Pennsylvania, United States of America, 4 Bureau of Tuberculosis Control, New York State Department of Health, Albany, New York, United States of America, 5 Colorado Department of Public Health and Environment, Disease Control and Environmental Epidemiology Division, Refugee Health Program, Denver, Colorado, United States of America, 6 Office of Refugee Health, Center for Infectious Diseases, California Department of Public Health, Sacramento, California, United States of America, 7 Texas Department of State Health Services, Austin, Texas, United States of America, 8 Marion County Public Health Department, Indianapolis, Indiana, United States of America, 9 Illinois Department of Public Health, Refugee Health Program, Chicago, Illinois, United States of America, 10 University of Louisville Division of Infectious Diseases, Louisville, Kentucky, United States of America

¤a Current address: Moravian College, Bethlehem, Pennsylvania, United States of America

¤b Current address: US Committee for Refugees and Immigrants, Austin, Texas, United States of America ¤c Current address: Illinois Department of Healthcare and Family Services, Chicago, Illinois, United States of America

¤d Current address: Kentucky Office for Refugees, Louisville, Kentucky, United States of America * wiz3@cdc.gov

Abstract

Background

Since 2008, the United States has issued between 2,000 and 19,000 Special Immigrant Visas (SIV) annually, with the majority issued to applicants from Iraq and Afghanistan. SIV holders (SIVH) are applicants who were employed by, or on behalf of, the US government or the US military. There is limited information about health conditions in SIV populations to help guide US clinicians caring for SIVH. Thus, we sought to describe health characteristics of recently arrived SIVH from Iraq and Afghanistan who were seen for domestic medical examinations.

Methods and findings

This cross-sectional analysis included data from Iraqi and Afghan SIVH who received a domestic medical examination from January 2014 to December 2016. Data were gathered

cooperative agreement CK12-1205 Strengthening Surveillance for Diseases among Newly Arrived Immigrants and Refugees. The CDC-funded nonresearch cooperative agreement CK12-1205 was intended to support states in improving their ability to collect data from the domestic medical examination for the purposes of public health surveillance and program evaluation. All decisions to collect and analyze data and draft manuscripts were made mutually through collaboration between CDC and the cooperative agreement participants. The nature of this cooperative agreement, as with other CDC cooperative agreements, was for the CDC program staff and the cooperative agreement participants to discuss and collaborate on all the agreement's strategies and activities.

Competing interests: The authors have declared that no competing interests exist.

Abbreviations: anti-HBc, antibody to hepatitis B core antigen; anti-HBs, antibody to hepatitis B surface antigen; CDC, Centers for Disease Control and Prevention; Cl, confidence interval; DS, Department of State; EBLL, elevated blood lead levels; EDN, Electronic Disease Notification; HBsAg, hepatitis B virus surface antigen; HBV, hepatitis B virus; HCV, hepatitis C virus; IRB, Institutional Review Board; LTBI, latent tuberculosis infection; PR, prevalence ratio; SD, standard deviation; SIV, Special Immigrant Visa; SIVH, SIV holder; STROBE, Strengthening the Reporting of Observational Studies in Epidemiology; TB, tuberculosis; TST, tuberculin skin test; USRAP, US Refugee Admissions Program.

from state refugee health programs in seven states (California, Colorado, Illinois, Kentucky, Minnesota, New York, and Texas), one county, and one academic medical center and included 6,124 adults and 4,814 children. Data were collected for communicable diseases commonly screened for during the exam, including tuberculosis (TB), hepatitis B, hepatitis C, malaria, strongyloidiasis, schistosomiasis, other intestinal parasites, syphilis, gonorrhea, chlamydia, and human immunodeficiency virus, as well as elevated blood lead levels (EBLL). We investigated the frequency and proportion of diseases and whether there were any differences in selected disease prevalence in SIVH from Irag compared to SIVH from Afghanistan. A majority of SIV adults were male (Iraqi 54.0%, Afghan 58.6%) and aged 18-44 (Iraqi 86.0%, Afghan 97.7%). More SIV children were male (Iraqi 56.2%, Afghan 52.2%) and aged 6-17 (Iragi 50.2%, Afghan 40.7%). The average age of adults was 29.7 years, and the average age for children was 5.6 years. Among SIV adults, 14.4% were diagnosed with latent tuberculosis infection (LTBI), 63.5% were susceptible to hepatitis B virus (HBV) infection, and 31.0% had at least one intestinal parasite. Afghan adults were more likely to have LTBI (prevalence ratio [PR]: 2.0; 95% confidence interval [CI] 1.5-2.7) and to be infected with HBV (PR: 4.6; 95% CI 3.6-6.0) than Iraqi adults. Among SIV children, 26.7% were susceptible to HBV infection, 22.1% had at least one intestinal parasite, and 50.1% had EBLL (\geq 5 mcg/dL). Afghan children were more likely to have a pathogenic intestinal parasite (PR: 2.7; 95% CI 2.4–3.2) and EBLL (PR: 2.0; 95% CI 1.5–2.5) than Iragi children. Limitations of the analysis included lack of uniform health screening data collection across all nine sites and possible misclassification by clinicians of Iraqi and Afghan SIVH as Iraqi and Afghan refugees, respectively.

Conclusion

In this analysis, we observed that 14% of SIV adults had LTBI, 27% of SIVH had at least one intestinal parasite, and about half of SIV children had EBLL. Most adults were susceptible to HBV. In general, prevalence of infection was higher for most conditions among Afghan SIVH compared to Iraqi SIVH. The Centers for Disease Control and Prevention (CDC) *Guidelines for the US Domestic Medical Examination for Newly Arriving Refugees* can assist state public health departments and clinicians in the care of SIVH during the domestic medical examination. Future analyses can explore other aspects of health among resettled SIV populations, including noncommunicable diseases and vaccination coverage.

Author summary

Why was this study done?

- Since 2008, the US has typically issued between 2,000 and 19,000 Special Immigrant Visas (SIV) annually, the majority to applicants from Iraq and Afghanistan.
- SIV holders (SIVH) are eligible for many of the same benefits as refugees under the US Refugee Admissions Program, including a domestic medical examination within 90 days of US arrival.

• Information about the health of SIV populations is limited and would be beneficial for US clinicians who see SIVH in their clinics.

What did the researchers do and find?

- In this cross-sectional analysis, data were gathered from domestic medical examinations of 10,938 SIVH from nine sites for diseases commonly screened for during the exam.
- We observed that 14.4% of SIV adults had LTBI, 26.9% of SIVH had at least one intestinal parasite, and about half of SIV children had elevated blood lead levels (EBLL). Most adults were susceptible to hepatitis B virus (HBV).
- In general, prevalence of infection was higher for most conditions among Afghan SIVH compared to Iraqi SIVH.

What do these findings mean?

- CDC's *Guidelines for the US Domestic Medical Examination for Newly Arriving Refugees* can assist state public health departments and clinicians in the care of SIVH during the routine domestic medical examination.
- Clinicians can ensure that SIVH receive appropriate screening and follow-up, paying particular attention to Afghan SIVH given their increased risk of several communicable diseases and EBLL (among children).
- Limitations of the analysis included lack of uniform health screening data collection across all nine sites and possible misclassification by clinicians of Iraqi and Afghan SIVH as Iraqi and Afghan refugees, respectively.

Introduction

Since 2008, the US has issued between 2,000 and 19,000 Special Immigrant Visas (SIV) annually, the majority to applicants from Iraq and Afghanistan [1]. SIV holders (SIVH) are applicants who were employed by, or on behalf of, the US government or the US military—notably as translators and interpreters with the US Armed Forces—and can include their immediate families [2]. Like refugees and other immigrants, all SIVH are required to receive a medical screening exam overseas according to the Technical Instructions written by the Centers for Disease Control and Prevention (CDC) [3]. SIVH are eligible for many of the same benefits as refugees under the US Refugee Admissions Program (USRAP); these benefits may include resettlement assistance and health-related benefits after arrival in the US [4,5]. SIVH have the option to receive these benefits before or after arrival. In addition, the CDC also recommends that SIVH and refugees receive a domestic medical examination within 90 days after arrival to the US [6]. However, SIVH do not have access to the overseas health interventions available to refugees, such as vaccines or parasite treatments, before departure to the US. Furthermore, SIVH must meet the requirements for vaccinations in CDC's Vaccination Technical Instructions for immigrants, whereas refugees may be granted a waiver for vaccinations [7]. Almost 80,000 SIVH received USRAP benefits from 2008 to 2018, including close to 18,500 from Iraq and 58,000 from Afghanistan [1]. There is little information about health conditions in SIV populations despite the high number who have elected USRAP benefits. Multiple studies have reported on the health of Iraqi and Afghan refugee populations [8–11]. However, SIVH encounter different circumstances while living abroad compared to refugees. SIVH are more likely to live in urban settings as opposed to camps before immigration and therefore likely to encounter different risks of disease. SIVH have faced threats to themselves and their families from militant groups while serving in roles such as interpreters or translators for the US government or military and often have gone into hiding [12, 13]. Therefore, it is possible that SIVH had less access to clinical and preventive health services, despite having the financial capacity to pay for these services.

Clinicians should follow CDC's Guidelines for the US Domestic Medical Examination for Newly Arrived Refugees during the domestic medical examination for SIVH [14]. They should (1) review SIVH's Department of State (DS) overseas medical examination forms, (2) obtain a complete medical history, (3) conduct a physical examination, (4) screen for communicable health conditions, such as tuberculosis (TB), (5) identify other conditions that may adversely affect resettlement or require further care, such as noncommunicable diseases (including mental health concerns), (6) provide preventive health interventions, such as immunizations, and (7) link the SIVH with appropriate care and follow-up [14]. The CDC guidelines are informed by disease epidemiology of the country of birth and country of residence prior to US resettlement. However, each state or community can adapt the screening guidelines to suit local circumstances and resources. Increasing clinician knowledge about common health conditions encountered in SIVH, including differences between Iraqi and Afghan SIVH, may facilitate diagnostic screening, physical examination, and referrals to additional healthcare providers in the US. To the authors' knowledge, this information is not currently available. Thus, we sought to describe health characteristics of recently arrived SIVH from Iraq and Afghanistan who were seen for domestic medical examinations. Specifically, we investigated the frequency and proportion of diseases commonly screened for in the domestic medical examination and, for certain diseases, whether there were any differences in prevalence for SIVH from Iraq compared to SIVH from Afghanistan.

Methods

Analysis population and data collection

This cross-sectional analysis included data from Iraqi and Afghan SIVH who received a domestic medical examination between January 2014 and December 2016. CDC collaborated with nine partners as part of a cooperative agreement (CK12-1205) to collect domestic medical examination data for analysis. Partners included the state refugee health programs in California, Colorado, Illinois, Kentucky, Minnesota, New York, and Texas; local partners in Marion County, Indiana; and an academic health center in Philadelphia, Pennsylvania. Data were obtained from partner refugee health databases. Some partners obtained individual-specific data from external programs (e.g., TB or lead prevention) in their jurisdictions. As part of a larger data collection effort, this analysis was not guided by a specific prospective analysis plan. However, the variables collected for analysis were outlined in a protocol used for ethical determination and shared with site partners contributing data. After data collection, we assessed the quality and availability of our data and identified analyses appropriate for our data, including this analysis describing health of SIVH. This analysis was determined to be non-research surveillance by a CDC human subjects advisor, and therefore formal Institutional Review Board

(IRB) review was not required. This analysis is reported as per the Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) guideline (S1 STROBE Checklist).

Demographic information (sex, age, and primary language spoken by the SIVH or used by the interpreter during the exam), nationality, and country of last residence were either provided by the site or, if a unique identifier was provided in the data set, obtained from matched records in CDC's Electronic Disease Notification (EDN) system. EDN is a centralized reporting system that notifies US state and local health departments and screening clinics of the arrival of immigrants with health conditions requiring medical follow-up, including TBrelated conditions, and all refugees. Sites also reported timing of the domestic medical examination relative to US arrival and whether a particular health condition was screened for. Screening test results were collected for TB, hepatitis B, hepatitis C, malaria, strongyloidiasis, schistosomiasis, presence of other intestinal parasites, syphilis, gonorrhea, chlamydia, HIV, and elevated blood lead levels (EBLL). Data were collected for these conditions, as they are recommended for screening according to CDC's Guidelines for the US Domestic Medical Examination for Newly Arriving Refugees [14]. For most conditions, we were unable to collect detailed information on the method of screening used by partners. We assumed the sites conducted screening or testing according to the CDC guidelines but recognize that there may be some variation by location, as the guidelines are meant to be customized in each jurisdiction [14]. While a comprehensive assessment of health of SIVH would include other conditions such as chronic diseases, screening for these conditions are based on clinical judgement and/ or state/clinic-based policies; hence, they are not routinely screened for during the domestic medical examination. Sites did not share these data.

For TB, information on diagnosis was reported and categorized as no evidence of TB, clinically active TB disease (person had clinical, bacteriological, and/or radiographic evidence of current pulmonary TB), not clinically active TB disease (person with a history of previous episode[s] of TB or abnormal stable radiographic findings and had a positive reaction to tuberculin skin test [TST], negative cultures, and no clinical and/or radiographic evidence of current disease), or latent tuberculosis infection (LTBI; person had a positive interferon gamma release assay or TST and negative diagnostic workup for TB) [15]. For hepatitis B, hepatitis B virus surface antigen (HBsAg), antibody to hepatitis B core antigen (anti-HBc), and antibody to hepatitis B surface antigen (anti-HBs) results were used to categorize a person's hepatitis B virus (HBV) infection status. Status was categorized as susceptible (HBsAg, anti-HBc, and anti-HBs all negative), uninfected/susceptibility unknown (HBsAg negative, anti-HBc and anti-HBs unknown), infected (HBsAg positive), immune through natural infection (HBsAg negative, anti-HBc positive, and anti-HBs positive), immune through hepatitis B (hepB) vaccination (HBsAg negative, anti-HBc negative, and anti-HBs positive), and immune but not specified (HBsAg negative, anti-HBs positive, and anti-HBc unknown) [16]. During the time of data collection, CDC guidelines recommended screening blood for lead in children ages 6 months to 16 years, with EBLL defined as \geq 5 mcg/dL [17]. For hepatitis C, malaria, strongyloidiasis, schistosomiasis, syphilis, gonorrhea, chlamydia, and HIV, the outcome was categorized as either "screened and positive" or "screened and negative." Persons who were screened and had unknown results reported were excluded from the analysis.

Data analysis

Frequencies and proportions were calculated to describe demographic characteristics and disease prevalence; results were stratified by SIV population (Afghan versus Iraqi) and age at screening visit (adult \geq 18 years, child <18 years). We used χ^2 tests to compare time of domestic medical examination and each disease condition by nationality of SIVH (Iraqi or Afghan). χ^2 tests were also used to determine any differences in disease conditions between SIVH and refugees. *p*-Values were reported if frequency per cell was \geq 5. Statistical significance was noted at a *p*-value <0.05. Denominators varied because of missing data and screening differences across sites. That is, nine sites provided screening data for TB, hepatitis B, hepatitis C, and lead; eight sites provided screening data for malaria, schistosomiasis, strongyloidiasis, intestinal parasites, and sexually transmitted infections. Thus, prevalence of health conditions was reported among those with available screening results. When calculating the proportions of persons who screened positive for a condition, we excluded sites that were only able to provide positive testing results without providing denominator data on how many people were screened for the condition, or who did not screen for or report a specific condition.

A modified Poisson regression was used to model the adjusted prevalence ratio (PR) (adjusting for age and sex) while accounting for state-level clustering. Nationality (Iraqi SIVH as reference) was the primary exposure variable, and LTBI, hepatitis B outcomes (susceptible, infected, and immune because of hepB vaccination), presence of at least one pathogenic parasite, and EBLL were the primary outcome variables. Medical conditions with five or fewer cases were excluded from the analyses. The prevalence of demographic characteristics and diseases among Iraqi and Afghan SIVH were compared to those of Iraqi and Afghan refugees (from same data source and time period) using χ^2 analysis. A *p*-value of <0.05 was defined as statistically significant.

Results

Of the 6,124 SIV adults included in our analysis, 1,112 (18.2%) were Iraqi and 5,012 (81.8%) were Afghan (Table 1). Of the 4,816 SIV children included in our analysis, 851 (17.7%) were Iraqi and 3,965 (82.3%) were Afghan. A majority of SIV adults were male (Iraqi 54.0%, Afghan 58.6%) and aged 18–44 (Iraqi 86.0%, Afghan 97.7%). More SIV children were male (Iraqi 56.2%, Afghan 52.2%) and aged 6–17 (Iraqi 50.2%, Afghan 40.7%). The average age of adults was 29.7 years (standard deviation [SD]: 7.6), and the average age for children was 5.6 years (SD: 4.4). The primary language spoken by or used by an interpreter for Iraqi SIVH was Arabic, and the primary languages spoken by or used by an interpreter for Afghan SIVH were Dari, Pashto, and Farsi. Before resettlement in the US, the majority of Iraqi and Afghan SIVH lived in their country of birth, while some Iraqis lived in Turkey or Jordan. About 97.4% of SIV adults and children had their domestic medical examination within 90 days of arrival in the US; this did not differ by nationality (Table 2).

SIV adults

Overall, 2,963 (85.4%) SIV adults had no evidence of TB, and 14.4% were diagnosed with LTBI (Table 2). Afghan adults were more likely to have LTBI compared to Iraqi adults (PR: 2.0; 95% confidence interval [CI] 1.5–2.7) (Table 3). Approximately 63.5% of adults were susceptible to HBV infection (Iraqi 72.2%, Afghan 61.5%) (Table 2). However, 10.7% of all adults were uninfected with HBV with unknown susceptibility (Iraqi 15.5%, Afghan 9.6%); therefore, the proportion susceptible to hepatitis B may be higher. Afghan adults were less likely to be susceptible to HBV infection compared to Iraqi adults (PR: 0.8; 95% CI 0.7–0.8) (Table 3). About 1.8% of all SIV adults were infected with HBV (Iraqi 0.6%, Afghan 2.1%), while 23.9% were immune (Iraqi 11.7%, Afghan 26.8%) (Table 2). Afghan adults were more likely to be infected with HBV (PR: 4.6; 95% CI 3.6–6.0) but were also more likely to have vaccine-induced immunity to HBV infection (PR: 2.0; 95% CI 1.5–2.6) compared to Iraqi adults (Table 3). Three percent of Iraqi and Afghan SIV adults screened positive for *Strongyloides* (Table 2). Thirty-one percent of screened adults had at least one intestinal parasite. *Blastocystis* (10.9%),

Demographic Characteristics		Adults \geq 18 years old	d	Children <18 years old			
	All Iraq A		Afghanistan	All	Iraq	Afghanistan	
	n (%)	n (%)	n (%)	n (%)	n (%)	n (%)	
Total	6,124	1,112 (18.2)	5,012 (81.8)	4,816	851 (17.7)	3,965 (82.3)	
Sex							
Female	2,587 (42.3)	512 (46.0)	2,075 (41.4)	2,267 (47.1)	372 (43.7)	1,895 (47.8)	
Male	3,537 (57.8)	600 (54.0)	2,937 (58.6)	2,548 (52.9)	478 (56.2)	2,070 (52.2)	
Age range (years)							
0-2				1,463 (30.4)	198 (23.3)	1,265 (31.9)	
3-5				1,313 (27.3)	226 (26.6)	1,087 (27.4)	
6–17				2,040 (42.4)	427 (50.2)	1,613 (40.7)	
18-44	5,850 (95.5)	956 (86.0)	4,894 (97.7)				
45-64	258 (4.2)	145 (13.0)	113 (2.3)				
≥65	16 (0.3)	11 (1.0)	5 (0.1)				
Primary language							
Dari	2,519 (41.1)	3 (0.3)	2,516 (50.2)	2,098 (43.6)	3 (0.4)	2,095 (52.8)	
Pashto	874 (14.3)	0 (0)	874 (17.4)	947 (19.7)	0 (0)	947 (23.9)	
Farsi	820 (13.4)	0 (0)	820 (16.4)	622 (12.9)	0 (0)	622 (12.9)	
Arabic	889 (14.5)	843 (75.8)	46 (0.9)	720 (15)	703 (82.6)	17 (0.4)	
Others ^b	1,036 (16.9)	266 (23.9)	770 (15.3)	431 (9)	145 (17.1)	286 (7.2)	
Country of last residence ^c							
Afghanistan	3,849 (62.9)	3 (0.3)	3,846 (76.7)	3,293 (68.4)	0 (0)	3,293 (83.1)	
Iraq	852 (13.9)	852 (76.6)	0 (0)	646 (13.4)	645 (75.8)	1 (0.03)	
Turkey	56 (0.9)	50 (4.5)	6 (0.1)	42 (0.9)	39 (4.6)	3 (0.1)	
Pakistan	30 (0.5)	0 (0)	30 (0.6)	11 (0.2)	0 (0)	11 (0.3)	
Jordan	24 (0.4)	24 (2.2)	0 (0)	17 (0.4)	17 (2.0)	0 (0)	
Others	66 (1.0)	14 (1.2)	49 (0.1)	39 (0.8)	15 (1.7)	23 (0.6)	
Unknown	1,247 (20.4)	166 (14.9)	1,081 (21.6)	768 (15.9)	134 (15.8)	634 (16.0)	

^aPercentages may not add up to 100% because of rounding.

^bOther languages include Uzbek, Kurdish, Armenian, English, and Urdu.

^cA total of 1,247 SIV adults (Iraqi SIV: 166 [14.9%]; Afghan SIV: 1,081 [21.6%]) had missing information about country of last residence. A total of 768 SIV children (Iraqi SIV: 134 [15.8%]; Afghan SIV: 634 [16.0%]) had missing information about country of last residence.

Abbreviation: SIV, Special Immigrant Visa

https://doi.org/10.1371/journal.pmed.1003083.t001

Giardia (2.8%), and *Dientamoeba* (2.7%) were the most common intestinal parasites found in both SIV populations. There was no difference in the prevalence of pathogenic intestinal parasites (excluding *Blastocystis* and *Dientamoeba*, which are considered controversial for treatment) between Afghan and Iraqi adults (PR: 1.6; 95% CI 1.0–2.6) (Table 3).

SIV children

Overall, 2,591 (96.9%) SIV children had no evidence of TB, and 3.1% had a diagnosis of LTBI (Table 2). Approximately 0.6% of all children were HBV infected, while 39.1% were immune (Iraqi 42.9%, Afghan 38.2%). Furthermore, about 26.7% of children were susceptible to HBV infection, although this proportion may be higher given that 33.6% of children were uninfected with their susceptibility unknown. Approximately 2.1% of children screened positive for *Strongyloides* infection, and 22.1% of children had at least one intestinal parasite. *Giardia* (8.1%), *Blastocystis* (7.0%), and *Dientamoeba* (2.3%) were the most common intestinal parasites

Health Screening Results	Adults ≥18 years old				Children <18 years old			
	All	Iraq	Afghanistan	<i>p</i> -value	All Iraq		Afghanistan	<i>p</i> -value
	n (%)	n (%)	n (%)		n (%)	n (%)	n (%)	
Total	6,124	1,112 (18.2)	5,012 (81.8)		4,816	851 (17.7)	3,965 (82.3)	
Time of domestic medical examination	n = 6063	<i>n</i> = 1,104	n = 4,959	0.54	n = 4,774	n = 843	<i>n</i> = 3,931	0.08
<30 days	3,281 (54.1)	580 (52.5)	2,701 (54.5)		2,764 (57.9)	466 (55.3)	2,298 (58.5)	
30-90 days	2,665 (44.6)	503 (45.6)	2,162 (43.6)		1,940 (40.6)	370 (43.9)	1,570 (39.9)	
>90 days	117 (1.9)	21 (1.9)	96 (1.9)		70 (1.5)	7 (0.8)	63 (1.6)	
TB ^d	<i>n</i> = 3471	<i>n</i> = 451	<i>n</i> = 3,020	0.01	n = 2,673	n = 320	n = 2,353	0.95
No evidence of TB	2,963 (85.4)	408 (90.7)	2,555 (84.6)		2,591 (96.9)	310 (96.9)	2,281 (96.9)	
Clinically active	2 (0.1)	0 (0)	2 (0.1)		0 (0)	0 (0)	0 (0)	
Not clinically active	8 (0.2)	1 (0.2)	7 (0.2)		0 (0)	0 (0)	0 (0)	
LTBI	498 (14.4)	42 (9.3)	456 (15.1)		82 (3.1)	10 (3.1)	72 (3.2)	
Hepatitis B ^e	n = 5777	<i>n</i> = 1,069	<i>n</i> = 4,708	< 0.0001	n = 3,988	<i>n</i> = 741	<i>n</i> = 3,247	< 0.0001
Susceptible	3,669 (63.5)	772 (72.2)	2,897 (61.5)		1,064 (26.7)	223 (30.1)	841 (25.9)	
Uninfected, susceptibility unknown	616 (10.7)	166 (15.5)	450 (9.6)		1,340 (33.6)	199 (26.9)	1,141 (35.1)	
Infected	106 (1.8)	6 (0.6)	100 (2.1)		23 (0.6)	1 (0.1)	22 (0.7)	
Immune								
Natural infection	331 (5.7)	15 (1.4)	316 (6.7)		33 (0.8)	3 (0.4)	30 (0.9)	
Hepatitis B vaccination	978 (16.9)	104 (9.7)	874 (18.6)		1,487 (37.3)	311 (42.0)	1,176 (36.2)	
Not specified	77 (1.3)	6 (0.6)	71 (1.5)		41 (1)	4 (0.5)	37 (1.1)	
Hepatitis C ^f	n = 4,007	n = 598	<i>n</i> = 3,409	N/A	n = 2,554	n = 342	n = 2,212	N/A
Screened, positive	31 (0.8)	4 (0.7)	27 (0.8)		8 (0.3)	2 (0.6)	6 (0.3)	
Malaria ^g	n = 1,572	n = 136	<i>n</i> = 1,436		<i>n</i> = 1,219	n = 83	<i>n</i> = 1,136	
Screened, positive	2 (0.1)	0 (0)	2 (0.1)		5 (0.4)	0 (0)	5 (0.4)	
Strongyloidiasis ^h	n = 2,578	<i>n</i> = 201	n = 2,377	0.6	<i>n</i> = 2,018	<i>n</i> = 131	<i>n</i> = 1,887	0.9
Screened, positive	75 (2.9)	7 (3.5)	68 (2.9)		50 (2.5)	3 (2.3)	47 (2.5)	
Schistosomiasis ^h	<i>n</i> = 130	<i>n</i> = 42	<i>n</i> = 88		<i>n</i> = 96	<i>n</i> = 17	n = 79	
Screened, positive	2 (1.5)	0 (0)	2 (2.3)		2 (2.1)	0 (0)	2 (2.5)	
Other intestinal parasites ⁱ	<i>n</i> = 3,982	n = 555	n = 3,427	0.3	<i>n</i> = 3,341	<i>n</i> = 451	n = 2,890	0.4
One or more intestinal parasites	1,235 (31.0)	183 (33.0)	1,052 (30.7)		737 (22.1)	93 (20.6)	644 (22.3)	
Blastocystis	435 (10.9)	46 (8.3)	389 (11.4)		233 (7.0)	22 (4.9)	211 (7.3)	
Giardia	111 (2.8)	10 (1.8)	101 (3.0)		272 (8.1)	17 (3.8)	255 (8.8)	
Dientamoeba	108 (2.7)	6 (1.1)	102 (3.0)		76 (2.3)	12 (2.7)	64 (2.2)	
Ascaris	60 (1.5)	0 (0)	60 (1.8)		45 (1.4)	0 (0)	45 (1.6)	
Entamoeba histolytica	74 (1.9)	9 (1.6)	65 (1.9)		25 (0.8)	2 (0.4)	23 (0.8)	
Hymenolepsis	3 (0.1)	0 (0)	3 (0.1)		17 (0.5)	0 (0)	17 (0.6)	
Others	2 (0.05)	0 (0)	2 (0.06)		5 (0.1)	0 (0)	5 (0.2)	
Syphilis ⁱ	<i>n</i> = 5,304	n = 928	<i>n</i> = 4,376	N/A	n = 723	<i>n</i> = 105	<i>n</i> = 618	N/A
Screened, positive	9 (0.2)	2 (0.2)	7 (0.2)		1 (0)	0 (0)	1 (0.2)	
Gonorrhea ^k	<i>n</i> = 18	<i>n</i> = 4	<i>n</i> = 14	N/A	<i>n</i> = 4	<i>n</i> = 2	<i>n</i> = 2	N/A
Screened, positive	0 (0)	0 (0)	0 (0)		0 (0)	0 (0)	0 (0)	
Chlamydia ^k	n = 1,573	n = 121	<i>n</i> = 1,452	N/A	<i>n</i> = 166	n = 15	n = 151	N/A
Screened, positive	14 (0.9)	2 (1.7)	12 (0.8)		2 (1.2)	0 (0)	2 (1.3)	
HIV ^k	n = 5,782	<i>n</i> = 1,027	n = 4,755	N/A	n = 3,924	n = 662	n = 3,262	N/A
Positive, screened/unscreened (type 1, type 2, or unknown)	4 (0.07)	1 (0.1)	3 (0.06)		1 (0.03)	1 (0.2)	0 (0)	
Blood lead level ¹				N/A	n = 2,368	n = 543	n = 1,825	< 0.0001
<5 mcg/dL	N/A	N/A	N/A		1,181 (49.9)	470 (86.6)	711 (39)	

Table 2. Domestic health screening results among adults and children who resettled to the US with Special Immigrant Visas (SIV), 2014–2016^{a,b,c}.

(Continued)

Table 2. (Continued)

Health Screening Results		Adults ≥18 years old			Children <18 years old			
	All	Iraq	Afghanistan	<i>p</i> -value	All	Iraq	Afghanistan	<i>p</i> -value
	n (%)	n (%)	n (%)		n (%)	n (%)	n (%)	
5–9 mcg/dL	N/A	N/A	N/A		944 (39.9)	71 (13.1)	873 (47.8)	
10–19 mcg/dL	N/A	N/A	N/A		201 (8.5)	2 (0.4)	199 (10.9)	
20-44 mcg/dL	N/A	N/A	N/A		36 (1.5)	0 (0)	36 (2)	
45–69 mcg/dL	N/A	N/A	N/A		2 (0.1)	0 (0)	2 (0.1)	
70+ mcg/dL	N/A	N/A	N/A		4 (0.2)	0 (0)	4 (0.2)	

^aPercentages may not add up to 100% because of rounding.

^bProportion of all SIVH who were not screened for a particular medical condition: latent TB (14.3%; 3% for Iraqi and Afghan adults, 35% for Iraqi children, and 27% for Afghan children); HBV (6.5%; 3% for Iraqi adults, 2% for Afghan adults, and 12% for Iraqi and Afghan children); hepatitis C virus (40.0%); malaria (74.4%); *Strongyloides* (57.9%); *Schistosoma* (97.9%), intestinal parasites (33.1%); syphilis (44.9%); gonorrhea (99.8%); chlamydia (84.1%); EBLL (77.5%); and HIV (17.6%). ^cWe used χ^2 tests to compare characteristic or disease condition by nationality (Iraqi or Afghan). *p*-Values were reported if frequency per cell was \geq 5. Statistical significance was noted at a *p*-value <0.05.

^dFor TB, information on diagnosis was reported and categorized as no evidence of TB, clinically active, not clinically active, and LTBI [15]. TB disease diagnosis was made by a positive smear, culture, or clinical diagnosis of pulmonary TB. A classification of not clinically active TB was made when a person had a history of previous episode(s) of TB or abnormal stable radiographic findings and had a positive reaction to TST, negative cultures, and no clinical and/or radiographic evidence of current disease. Diagnosis of LTBI was made by a positive interferon gamma release assay (IGRA) or TST and negative diagnostic workup for TB. The majority of adults (95%) were tested using IGRA. Among children tested for LTBI, 87% were tested using IGRA and 17% were tested using TST. Data were included if states provided information about TB diagnosis for an individual.

^eHBV status was categorized as susceptible (HBsAg, anti-HBc, and anti-HBs all negative), uninfected/susceptibility unknown (HBsAg negative, anti-HBc and anti-HBs unknown), infected (HBsAg positive), immune through natural infection (HBsAg negative, anti-HBc positive, and anti-HBs positive), immune through hepatitis B vaccination (HBsAg negative, anti-HBc negative, and anti-HBs positive) and immune but not specified (HBsAg negative, anti-HBs positive, and anti-HBc unknown) [16]. About 3.5% of all SIV adults and 6% of all SIV children had HBV results where the interpretation was unknown or unclear.

^fHepatitis C was diagnosed by any of the following: detection of antibody to hepatitis C virus (anti-HCV), a positive recombinant immunoblot assay (RIBA) result, or a positive HCV RNA polymerase chain reaction (PCR) result.

^gMalaria diagnosis was laboratory confirmed using either microscopy or by a rapid diagnostic test.

^hStrongyloides and Schistosoma diagnoses were laboratory confirmed using either microscopy or by serology testing.

ⁱIntestinal parasite infection diagnoses were laboratory confirmed using stool ova and parasite testing. Reported *p*-value compares Iraqi and Afghan SIVH with any intestinal parasite reported (pathogenic and nonpathogenic) to Iraqi and Afghan SIVH with no intestinal parasite reported.

^jSyphilis diagnosis was made via a positive non-treponemal test (venereal disease research laboratory [VDRL] or rapid plasma reagin [RPR]) followed by a positive confirmatory treponemal test (e.g., *Treponema pallidum*-particle agglutination [TP-PA], microhemagglutination assay for *T. pallidum* [MHA-TP]). Syphilis testing is recommended in all persons \geq 15 years of age if no overseas testing results are available, and in persons <15 years of age if sexually active.

^kGonorrhea, chlamydia, and HIV diagnoses were made via laboratory-confirmed testing.

¹Blood lead level screening applies to children from 6 months up to 16 years of age only.

Abbreviations: anti-HBc, antibody to hepatitis B core antigen; anti-HBs, antibody to hepatitis B surface antigen; EBLL, elevated blood lead levels; HBsAg, hepatitis B virus surface antigen; HBV, hepatitis B virus; HCV, hepatitis C virus; LTBI, latent tuberculosis infection; SIV, Special Immigrant Visa; SIVH, SIV holder; TB, tuberculosis; TST, tuberculin skin test

https://doi.org/10.1371/journal.pmed.1003083.t002

found. Afghan children were more likely to have pathogenic intestinal parasites (excluding *Blastocystis* and *Dientamoeba*) compared to Iraqi children (PR: 2.7; 95% CI 2.4–3.2) (Table 3). Over half of all SIV children had an EBLL, with higher prevalence among Afghans than among Iraqis (PR: 2.0; 95% CI 1.5–2.5).

SIVH versus refugees

Iraqi and Afghan SIVH were compared to refugees with the same nationality (Iraqi 9,368; Afghan 1,407). Afghan SIV adults (17.8%) were less likely to have LTBI than were Afghan refugee adults (26.1%) ($\chi^2 p < 0.0001$), and Afghan SIV children (39.9%) were less likely to be

Medical Conditions	Adults \geq 18 years old	Children <18 years old		
	aPR (95% CI) Ref: Iraqi SIVH	aPR (95% CI) Ref: Iraqi SIVH		
LTBI	2.0 (1.5–2.7)	1.0 (0.4–2.7)		
Hepatitis B ^b				
Susceptible	0.8 (0.7–0.8)	1.0 (0.7–1.3)		
Infected	4.6 (3.5–6.0)	N/A		
Immune through vaccination	2.0 (1.5–2.6)	1.0 (0.8–1.2)		
Pathogenic parasites	1.6 (1.0–2.6)	2.7 (2.4–3.2)		
EBLL (\geq 5 mcg/dL)	N/A	2.0 (1.5–2.5)		

Table 3. Adjusted PRs for select medical conditions comparing Afghan and Iraqi Special Immigrant Visa Holders (SIVH) who resettled to the US, 2014–2016^a.

^aPoisson regression was used to model the adjusted PRs (adjusted for age and sex) to assess association of nationality and outcomes. Iraqi nationality was used as reference.

^bHBV status was categorized as susceptible (HBsAg, anti-HBc, and anti-HBs all negative), infected (HBsAg positive), and immune through hepatitis B vaccination (HBsAg negative, anti-HBc negative, and anti-HBs positive). Abbreviations: anti-HBc, antibody to hepatitis B core antigen; anti-HBs, antibody to hepatitis B surface antigen; aPR, adjusted prevalence ratio; CI, confidence interval; EBLL, elevated blood lead levels; HBsAg, hepatitis B virus surface antigen; HBV, hepatitis B virus; LTBI, latent tuberculosis infection; Ref, reference; SIVH, SIV holder

https://doi.org/10.1371/journal.pmed.1003083.t003

susceptible to HBV infection compared to Afghan refugee children (52.8%) ($\chi^2 p < 0.0001$). Afghan SIV adults (30.7%) and children (22.3%) had higher prevalence of at least one intestinal parasite (other than *Strongyloides* or schistosomiasis) than their respective Afghan refugee populations (adults 12.5%, children 13.4%) ($\chi^2 p < 0.0001$). Iraqi SIV children (41.1%) were less likely to be susceptible to hepatitis B than were Iraqi refugee children (47.2%) ($\chi^2 p = 0.01$). Iraqi SIV adults (33.0%) had higher prevalence of at least one intestinal parasite relative to Iraqi refugee adults (25.9%) ($\chi^2 p = 0.0004$). The prevalence of at least one intestinal parasite and EBLL did not differ between SIV and refugee children from both Iraq and Afghanistan.

SIVH not screened for a particular condition

The proportions of all SIVH who were not screened for a particular medical condition are LTBI (14.3%), HBV (6.5%), hepatitis C virus (HCV) (40.0%), malaria (74.4%), *Strongyloides* (57.9%), *Schistosoma* (97.9%), intestinal parasites (33.1%), syphilis (44.9%), gonorrhea (99.8%), chlamydia (84.1%), EBLL (77.5%), and HIV (17.6%). These are crude estimates only and not adjusted for age, sex, or nationality.

Discussion

In this analysis, we described the health of SIV populations after arrival into the US, focusing on conditions typically assessed at the domestic medical examination according to CDC's *Guidelines for the US Domestic Medical Examination for Newly Arrived Refugees*. Key findings were reported on the health of 10,940 SIVH from Iraq and Afghanistan who received the domestic medical examination in nine regions in the US. First, about 14% of adults had LTBI, with a higher prevalence in Afghans than in Iraqis. Second, the majority of adults were susceptible to HBV infection, with Iraqis more likely than Afghans to be susceptible (i.e., no evidence of immunity due to prior infection or vaccination). Third, about 27% of SIVH had at least one intestinal parasite infection. Finally, over half of all SIV children had EBLL; the prevalence among Afghan children (61%) was higher than the prevalence among Iraqis.

While published data on LTBI prevalence among Afghan populations in general are limited, LTBI prevalence estimates in Iraqi populations range from 0.9% to 14.1% [8–10]. The incidence of TB disease in Afghanistan is 189 per 100,000 people compared to 42 per 100,000 people in Iraq [18,19], which may explain our finding of higher prevalence of LTBI among Afghans. Domestic clinicians screening newly arrived SIVH should refer to CDC's *Guidelines for the US Domestic Medical Examination for Newly Arrived Refugees* to ensure they are appropriately screening for and treating LTBI, with special attention to adults, who are not routinely tested for *Mycobacterium tuberculosis* infection in the overseas medical exam [20].

About 2% of SIV adults and 1% of SIV children presented with chronic HBV infection, while the majority of SIV adults and a quarter of SIV children were susceptible to HBV infection. Multiple studies have documented estimates of chronic HBV infection in predominantly refugee and migrant Iraqi populations ranging from 0.4% to 2.5% [8,10,21–24]. Chronic HBV infection estimates in Afghan populations range from 1.5% to 9.2% [23;25–30], with higher burden in refugees [25,27]. Other studies similarly showed that susceptibility to hepatitis B among resettled adult Iraqi populations was over 70% [23,31] and over 50% among adult Afghan populations [31].

While hepatitis B vaccination was introduced in Iraq in 1985 to all children and adults with high-risk conditions, routine vaccination has been interrupted by war and military action in the early part of the 21st century [32]. Through the WHO Expanded Program on Immunization, the hepatitis B vaccine birth dose was introduced in Iraq in 2004, and infants were eligible to receive all three doses of the vaccine [33]. Hepatitis B vaccination was introduced in Afghanistan in 2006, while the birth dose was introduced only in 2014 [33]. In 2017, WHO estimated that 63% of Iraqi infants and 65% of Afghan infants received all three doses of the vaccine [33]. It is unclear why at least a third of Iraqi children and a quarter of Afghan children in our analysis were susceptible to hepatitis B given that hepatitis B vaccination should be administered (or evidence of previous vaccination documented) to all SIV children up through age 18 via routine childhood immunization programs or during the overseas predeparture screening exam per the Vaccination Technical Instructions [7]. Possible explanations for hepatitis B susceptibility among SIV children could include lack of overseas vaccine availability, recent administration of the vaccine (leaving insufficient time for detectable antibody levels to develop), the more recent introduction of hepatitis B vaccine in Afghanistan, inadequate education of parents about vaccination importance, inappropriately timed or missed immunizations, and inaccurate immunization cards [34].

Similar to our analysis, other publications report that vaccine-induced immunity to hepatitis B is lower for the Iraqi adult population (5%) than for the Afghan population (40%) [31]. In Iraq, the hepatitis B vaccine is given in adulthood only to high-risk adults, so it is possible that the vaccine was not given to most Iraqi adults while they were in Iraq. Another explanation for the low prevalence of vaccine-induced immunity to HBV among both Iraqi and Afghan populations is that SIV families who go into hiding while serving the US government or military abroad may have limited access to healthcare and preventive health services [12,13].

Unlike refugees, SIVH are not currently eligible to participate in the voluntary Vaccination Program for US-bound Refugees, which was created to provide 1–2 doses of certain vaccines, including hepatitis B, overseas [35]. However, as mentioned previously, SIV children through age 18 applying for US residency are required to receive the first dose of the hepatitis B vaccine series to comply with immigration requirements [7], whereas SIV adults are unlikely to receive hepatitis B vaccine. The overseas vaccination program for refugees is variably implemented across different sites in the Middle East, which could possibly explain why SIV Iraqi and Afghan children are less susceptible to HBV infection compared to US-bound Iraqi and Afghan refugee children, who may have not received doses of the hepatitis B vaccine series.

Both Iraq and Afghanistan are countries with intermediate hepatitis B endemicity (2% to 7% of the population with chronic hepatitis B infection), and the prevalence of HBV infection susceptibility is high among SIVH. US clinicians caring for Iraqi and Afghan SIVH should continue hepatitis B screening for all adults and children, and offer vaccinations to those susceptible to infection, per CDC's *Guidelines for the US Domestic Medical Examination for Newly Arriving Refugees*: Screening for Viral Hepatitis During the Domestic Medical Examination of Newly Arrived Refugees [36,37].

The most common intestinal parasites seen in both SIV adults and children included *Blastocystis*, *Giardia*, and *Dientamoeba*. Studies in Iraqi populations report a high overall prevalence of gastrointestinal infections with estimates up to 64% of the population, and commonly reported parasites include *Blastocystis* (32%–36%) [8, 38], *Giardia* (28%–45%) [8,38–41], and *Entamoeba* (23%) [39]. In Afghan populations, overall prevalence of intestinal parasites as high as 39% has been noted [42], and commonly reported parasites include *Giardia* (15%– 59%) [42–44] and *Ascaris* (20%–35%) [41]. *Blastocystis* was found in 15% of Polish soldiers returning from peacekeeping missions in both Afghanistan and Iraq; the infections were presumably contracted in these regions [45]. Several parasites may not cause symptoms; however, others have the potential to cause complications. SIVH, unlike Iraqi and Afghan refugees, do not receive the predeparture parasite treatment as part of CDC's overseas presumptive treatment program. Therefore, US clinicians caring for SIVH during the domestic medical examination should strongly consider screening and treating for intestinal parasites according to CDC's *Guidelines for the US Domestic Medical Examination for Newly Arriving Refugees*: *Domestic Intestinal Parasite Guidelines* to prevent further transmission or complications [46].

Few SIVH from Iraq and Afghanistan were reported positive for any of the following: syphilis, gonorrhea, chlamydia, or HIV. However, the majority of SIVH were not screened for gonorrhea and chlamydia. The CDC's *Guidelines for the US Domestic Medical Examination for Newly Arriving Refugees*: Screening for Sexually Transmitted Diseases during the Domestic Medical Examination for Newly Arrived Refugees recommend screening for certain sexually transmitted infections based on whether overseas screening was done (for syphilis), symptoms, sitebased clinical screening policies for these conditions, age, and other risk factors [47].

Half of all SIV children had a blood lead level of 5 mcg/dL or greater, with 61% of Afghan children screening positive for EBLL and 13.2% recording a blood lead level above 10 mcg/dL. One study among resettled Afghan refugee children similarly revealed prevalence of EBLL up to 55% [48]. There have been few reports of lead exposure in Afghan children, and these identified the use of lead-containing eye cosmetics, a common cultural practice among even very young children and babies from Afghanistan and several Middle Eastern and African countries [48], as the source of lead exposure [49,50]. Studies among resettled Iraqi refugee children show that about 20% of children (versus 13% in our analysis) have blood lead levels over 5 mcg/dL [9, 49], while about 1% to 3% of children have levels above 10 mcg/dL [8,10,51]. While potential lead exposures among SIV children were not evaluated in our analysis, US clinicians should ensure that all SIV children 16 years of age and under are evaluated at the domestic medical examination and managed for lead poisoning and malnutrition (including an evaluation for iron deficiency) after arrival [17]. Management includes discussing potential lead exposures with parents of those children with higher levels and retesting blood lead levels of all children 6 years of age and under within 3–6 months after resettlement [17].

This analysis has some limitations. First, health screening data were not collected and reported uniformly across all nine sites; therefore, denominators across medical conditions and diagnoses vary. For example, *Blastocystis hominis* was reported as a pathogenic parasite by some clinicians and a nonpathogenic parasite by other clinicians (treatment for controversial parasites, such as *Blastocystis* and *Dientamoeba*, is considered if there are no other

explanations for symptoms [46]). To account for this variation, for states that diagnosed Blastocystis or Dientamoeba only in an SIVH but reported the parasite as pathogenic, we excluded these data in the calculation of the adjusted PR for pathogenic parasites. Second, the CDC's Guidelines for the US Domestic Medical Examination for Newly Arriving Refugees differentiate screening according to nationality, age of patient, and documented overseas test results and interventions; thus, not all tests were conducted for all SIVH. Third, it is possible that clinicians could misclassify Iraqi and Afghan SIVH as refugees. Hence, disease estimates could be underestimated or overestimated. Fourth, given that about 11% of SIV adults and 34% of SIV children were uninfected but had unknown susceptibility to HBV, the estimates for HBV susceptibility and immunity could be overestimated or underestimated. Fifth, sites only shared whether SIVH screened positive or negative for different infections such as HCV infection, but the test used was not reported. With HCV infection, it is possible that a person can test positive for HCV antibody but have undetectable levels of HCV RNA, indicating that a person may not be currently infected. With *Schistosoma*, a patient may test positive via serology but test negative via a stool (or urine) test, which also may not indicate current infection. Stool microscopy has a low sensitivity for Strongyloides and may not capture all cases. Another example is that Entamoeba histolytica cannot be easily distinguished from Entamoeba dispar, a nonpathogenic parasite, via stool microscopy. Therefore, based on the screening test used, the estimates for these infections reported in this article may be underestimated or overestimated. Sixth, there may be site-based differences that may impact screening outcomes that were not taken into account in our analysis. These differences include availability of transportation for SIVH, culturally competent translators, funding, and state-based health insurance policies. Lastly, given that data from SIVH who did not attend the domestic medical examination were not obtained, we were unable to determine if there were differences in disease estimates between SIVH who elected benefits versus those who did not, and those who attended the domestic medical examination versus those who did not.

To the authors' knowledge, this is the first analysis of select communicable diseases and blood lead levels from the domestic medical examination among Iraqi and Afghan SIV populations and one of few to explore the health of these populations after resettlement to the US. Given that sites collected and shared data differently, and we did not capture detailed information on method or type of testing for most conditions, our findings only reflect those sites contributing data and not all SIV populations in the US.

In conclusion, we observed that 14% of SIV adults had LTBI, 27% of SIVH had at least one intestinal parasite, and about half of SIV children had EBLL. Most adults were susceptible to HBV. In general, prevalence of infection was higher for most conditions among Afghan SIVH compared to Iraqi SIVH. CDC's Guidelines for the US Domestic Medical Examination for Newly Arrived Refugees can assist state public health departments and clinicians in the care of refugees and others receiving refugee benefits (including SIVH) during the routine domestic medical examination. Based on the results of this analysis, clinicians can ensure that SIVH receive the appropriate screening procedures and followed up with as necessary, paying particular attention to Afghan SIVH given their increased risk of several communicable diseases and EBLL (among children). Clinicians should also be aware that Iraqi and Afghan SIV populations differ in socioeconomic and migration histories from refugee populations from the same countries. Furthermore, access to overseas interventions and vaccination requirements differ between SIV and US-bound refugee populations. This could in part explain differences in disease estimates between SIV and refugee populations noted in this analysis. Future analyses can explore other aspects of health among resettled SIV populations, including noncommunicable diseases, vaccination coverage, mental health conditions, and long-term health outcomes.

Supporting information

S1 STROBE Checklist. STROBE, Strengthening the Reporting of Observational Studies in Epidemiology. (DOCX)

Acknowledgments

The authors would like to thank Allison Pauly (formerly with the Kentucky Office for Refugees) for her contributions to the project design and data acquisition from her program. The authors would also like to thank Kenneth Mulanya (formerly with Marion County, IN) and Shandy Dearth (Marion County, IN) for their contributions to data stewardship and direction of grant activities, respectively. The authors also wish to thank the Colorado Department of Public Health and Environment Disease Control and Environmental Epidemiology Division programs for data partnership and subject matter expertise, and Carol Tumaylle, MPH, Colorado State Refugee Health Coordinator, Colorado Department of Human Services, for the administration of the overall domestic health screening exam program for all refugees in Colorado. The authors would also like to thank Dr. Drew Posey, Dr. Noelle Nelson, and Dr. Heather Burke from the CDC for their subjective matter expertise and review of the manuscript.

Disclaimer: The findings and conclusions in this report are those of the authors and do not necessarily represent the official position of CDC.

Author Contributions

Conceptualization: Gayathri S. Kumar, Emily S. Jentes.

Data curation: Clelia Pezzi, Blain Mamo, Kevin Scott, Colleen Payton, Kailey Urban, Stephen Hughes, Lori Kennedy, Nuny Cabanting, Jessica Montour, Melissa Titus, Jenny Aguirre, Breanna Kawasaki, Rebecca Ford.

Formal analysis: Clelia Pezzi.

Investigation: Gayathri S. Kumar, Clelia Pezzi, Simone Wien, Blain Mamo, Kevin Scott, Colleen Payton, Kailey Urban, Stephen Hughes, Lori Kennedy, Nuny Cabanting, Jessica Montour, Melissa Titus, Jenny Aguirre, Breanna Kawasaki, Rebecca Ford, Emily S. Jentes.

Methodology: Gayathri S. Kumar, Clelia Pezzi, Simone Wien.

Project administration: Gayathri S. Kumar.

Software: Gayathri S. Kumar.

Supervision: Emily S. Jentes.

- Writing original draft: Gayathri S. Kumar.
- Writing review & editing: Clelia Pezzi, Simone Wien, Blain Mamo, Kevin Scott, Colleen Payton, Kailey Urban, Stephen Hughes, Lori Kennedy, Nuny Cabanting, Jessica Montour, Melissa Titus, Jenny Aguirre, Breanna Kawasaki, Rebecca Ford, Emily S. Jentes.

References

 US Department of State, Bureau of Population, Refugees, and Migration, Office of Admissions, Refugee Processing Center. Admissions & Arrivals, SIV Arrivals by Nationality by Month. [cited 2019 Mar 22]. Available from: https://www.wrapsnet.org/admissions-and-arrivals/.

- 2. US Department of State, Bureau of Consular Affairs. Directory of Visa Categories. [cited 2019 May 21]. Available from: https://travel.state.gov/content/travel/en/us-visas/visa-information-resources/all-visacategories.html.
- US Department of Health and Human Services, Centers for Disease Control and Prevention (CDC). Technical Instructions for Panel Physicians. [cited 2019 Apr 2019]. Available from: https://www.cdc. gov/immigrantrefugeehealth/exams/ti/panel/technical-instructions-panel-physicians.html.
- US Department of State, Bureau of Population, Refugees, and Migration. SIV Refugee Benefits Election Form 2017 (English). [cited 2019 May 21]. Available from: https://www.wrapsnet.org/siv-iraqisyrian-p-2/.
- US Department of Health and Human Services, Office of the Administration for Children & Families, Office of Refugee Resettlement. Health Insurance. [cited 2019 May 21]. Available from: https://www. acf.hhs.gov/orr/health.
- 6. US Department of Health and Human Services, Office of Refugee Resettlement. Revised Medical Screening Guidelines for Newly Arriving Refugees. [cited 2019 May 21]. Available from: http://www.acf. hhs.gov/sites/default/files/orr/state_letter_12_09_revised_medical_screening_guidelines_for_newly. pdf.
- US Department of Health and Human Services, CDC. Vaccination Technical Instructions for Panel Physicians. [cited 2019 Feb 20]. Available from: https://www.cdc.gov/immigrantrefugeehealth/exams/ti/ panel/vaccination-panel-technical-instructions.html.
- US Department of Health and Human Services, CDC. Health of resettled Iraqi refugees–San Diego County, California, October 2007–September 2009. MMWR Morb Mortal Wkly Rep. 2010; 59:1614– 1618. PMID: 21160458
- Yanni EA, Naoum M, Odeh N, Han P, Coleman M, Burke H. The health profile and chronic diseases comorbidities of US-bound Iraqi refugees screened by the International Organization for Migration in Jordan: 2007–2009. J Immigr Minor Health. 2013; 15:1–9. <u>https://doi.org/10.1007/s10903-012-9578-6</u> PMID: 22307545
- Yun K, Matheson J, Payton C, Scott KC, Stone BL, Song L, et al. Health profiles of newly arrived refugee children in the United States, 2006–2012. Am J Public Health. 2016; 106:128–135. https://doi.org/ 10.2105/AJPH.2015.302873 PMID: 26562126
- Sanati Pour M, Kumble S, Hanieh S, Biggs B. Prevalence of dyslipidemia and micronutrient deficiencies among newly arrived Afghan refugees in rural Australia: a cross-sectional study. BMC Public Health. 2014; 14:896. https://doi.org/10.1186/1471-2458-14-896 PMID: 25175525
- 12. Huetteman E. 'They Will Kill Us': Afghan Translators Plead for Delayed U.S. Visas. The New York Times. 2019 Aug 9. [cited 2019 Mar 19]. Available from: https://www.nytimes.com/2016/08/10/us/politics/afghan-translators-military-visas.html.
- 13. Al Jazeera. Afghan Translators: Out of Hiding. 2016 Nov 15. [cited 2019 Mar 19]. Available from: https:// www.aljazeera.com/programmes/faultlines/2016/11/afghan-translators-hiding-161114104958282. html.
- US Department of Health and Human Services, CDC. Guidelines for the US Domestic Medical Examination for Newly Arriving Refugees. [cited 2019 Feb 21]. Available from: <u>http://www.cdc.gov/</u> immigrantrefugeehealth/guidelines/domestic/domestic-guidelines.html.
- American Thoracic Society. Diagnostic Standards and Classification of Tuberculosis in Adults and Children. Am J Respir Crit Care Med. 2000; 161:1376–1395. https://doi.org/10.1164/ajrccm.161.4.16141 PMID: 10764337
- Mitruka K, Pezzi C, Baack B, Burke H, Cochran J, Matheson J, et al. Evaluation of hepatitis B virus screening, vaccination, and linkage to care among newly arrived refugees in four states, 2009–2011. J Immigr Minor Health. 2019; 21:39–46. https://doi.org/10.1007/s10903-018-0705-x PMID: 29417356
- US Department of Health and Human Services, CDC. Screening for Lead during the Domestic Medical Examination for Newly Arrived Refugees. [cited 2019 Nov 15]. Available from: https://www.cdc.gov/ immigrantrefugeehealth/guidelines/lead-guidelines.html.
- World Health Organization. Afghanistan: Tuberculosis profile. [cited 2019 Jul 29]. Available from: https://extranet.who.int/sree/Reports?op=Replet&name=/WHO_HQ_Reports/G2/PROD/EXT/ TBCountryProfile&ISO2=AF&outtype=html.
- World Health Organization. Iraq: Tuberculosis profile. [cited 2019 Jul 29]. Available from: https:// extranet.who.int/sree/Reports?op=Replet&name=/WHO_HQ_Reports/G2/PROD/EXT/ TBCountryProfile&ISO2=IQ&outtype=html.
- US Department of Health and Human Services, CDC. Guidelines for Screening for Tuberculosis Infection and Disease during the Domestic Medical Examination for Newly Arrived Refugees. [cited 2019 Jul 31]. Available from: https://www.cdc.gov/immigrantrefugeehealth/guidelines/domestic/tuberculosis-guidelines.html.

- Mixson-Hayden T, Lee D, Ganova-Raeva L, Drobeniuc J, Stauffer WM, Teshale E, et al. Hepatitis B virus and hepatitis C virus infections in United States-bound refugees from Asia and Africa. Am J Trop Med Hyg. 2014; 90:1014–1020. https://doi.org/10.4269/ajtmh.14-0068 PMID: 24732462
- Chironna M, Germinario C, Lopalco PL, Carrozzini F, Barbuti S, Quarto M. Prevalence rates of viral hepatitis infections in refugee Kurds from Iraq and Turkey. Infection. 2003; 31:70–74. <u>https://doi.org/10.1007/s15010-002-3100-3 PMID: 12682810</u>
- Richter C, Ter Beest G, Gisolf EH, Van Bentum P, Waegemaekers C, Swanink C, et al. Screening for chronic hepatitis B and C in migrants from Afghanistan, Iran, Iraq, the former Soviet Republics, and Vietnam in the Arnhem region, The Netherlands. Epidemiol Infect. 2014; 142:2140–2146. https://doi.org/ 10.1017/S0950268813003415 PMID: 24398373
- Scott KC, Taylor EM, Mamo B, Herr ND, Cronkright PJ, Yun K, et al. Hepatitis B screening and prevalence among resettled refugees—United States, 2006–2011. MMWR Morb Mortal Wkly Rep. 2015; 64:570–573. PMID: 26042647
- Khanani MR, Ansari AS, Kazmi SU, Ali SH. Concentrated epidemics of HIV, HCV, and HBV among Afghan refugees. J Infect. 2010; 61:434–437. https://doi.org/10.1016/j.jinf.2010.08.009 PMID: 20831883
- 26. Tanju IA, Levent F, Sezer RG, Cekmez F. Hepatitis B, hepatitis C, and human immunodeficiency virus seropositivity among children in Kabul, Afghanistan: A cross-sectional study. Hepat Mon. 2014; 14(3): e16154. https://doi.org/10.5812/hepatmon.16154 PMID: 24693318
- Nasir A, Todd CS, Stanekzi MR, Bautista CT, Botros BA, Scott PT, et al. Prevalence of HIV, hepatitis B and hepatitis C and associated risk behaviours amongst injecting drug users in three Afghan cities. Int J Drug Policy. 2011; 22:145–152. https://doi.org/10.1016/j.drugpo.2010.10.006 PMID: 21146392
- Khan S, Attaullah S. Share of Afghanistan populace in hepatitis B and hepatitis C infection's pool: is it worthwhile? Virol J. 2011; 8:216. https://doi.org/10.1186/1743-422X-8-216 PMID: 21569317
- 29. Todd CS, Abed AM, Strathdee SA, Scott PT, Botros BA, Safi N, et al. HIV, hepatitis C, and hepatitis B infections and associated risk behavior in injection drug users, Kabul Afghanistan. Emerg Infect Dis. 2007; 13:1327–1331. https://doi.org/10.3201/eid1309.070036 PMID: 18252103
- Todd CS, Nasir A, Stanekzai MR, Bautista CT, Botros BA, Scott PT, et al. HIV, hepatitis B, and hepatitis C prevalence and associated risk behaviors among female sex workers in three Afghan cities. AIDS. 2010; 24(suppl 2):S69–S75.
- Freidl GS, Tostmann A, Curvers M, Ruijs WLM, Smits G, Schepp R, et al. Immunity against measles, mumps, rubella, varicella, diphtheria, tetanus, polio, hepatitis A and hepatitis B among adult asylum seekers in the Netherlands, 2016. Vaccine. 2018; 36:1664–1672. https://doi.org/10.1016/j.vaccine. 2018.01.079 PMID: 29454516
- Centers for Disease Control and Prevention. Vaccination Services in Postwar Iraq, May 2003. MMWR Morb Mortal Wkly Rep. 2003; 52:734–735. PMID: 12904740
- World Health Organization. Immunization, Vaccines and Biologicals: Data, statistics and graphics. [cited 2019 Mar 1]. Available from: https://www.who.int/immunization/monitoring_surveillance/data/en/.
- **34.** Al-lela OQB, Bahari MB, Baderden SK, Basher AY, Hamoodi HK. Factors affecting on immunization compliance: Iraq. J Pharm Pract Community Med. 2017; 3:246–253.
- US Department of Health and Human Services, CDC. Vaccination Program for U.S.-Bound Refugees. [cited 2019 Mar 15]. Available from: https://www.cdc.gov/immigrantrefugeehealth/guidelines/overseas/ interventions/immunizations-schedules.html.
- **36.** US Department of Health and Human Services, CDC. Screening for Viral Hepatitis During the Domestic Medical Examination of Newly Arrived Refugees. [cited 2019 Jan 15]. Available from: https://www.cdc. gov/immigrantrefugeehealth/guidelines/domestic/hepatitis-screening-guidelines.html.
- US Department of Health and Human Services, CDC. Evaluating and Updating Immunizations during the Domestic Medical Examination for Newly Arrived Refugees. [cited 2019 Feb 10]. Available from: https://www.cdc.gov/immigrantrefugeehealth/guidelines/domestic/immunizations-guidelines.html.
- Mahdi NK, Ali NH. Intestinal parasites, including Cryptosporidium species, in Iraqi patients with sicklecell anaemia. East Mediterr Health J. 2002; 8:345–349. PMID: 15339123
- Al-Saeed AT, Issa SH. Frequency of Giardia lamblia among children in Dohuk, northern Iraq. East Mediterr Health J. 2006; 12:555–561. PMID: 17333793
- **40.** Mahdi NK, Setrak SK, Shiwaish SM. Diagnostic methods for intestinal parasites in southern Iraq with reference to *Strongyloides stercoralis*. Southeast Asian J Trop Med Public Health. 1993; 24:685–91. PMID: 7939941
- 41. Al-Kubaisy W, Al-Talib H, Al-khateeb A, Shanshal MM. Intestinal parasitic diarrhea among children in Baghdad—Iraq. Trop Biomed. 2014; 31:499–506. PMID: 25382477

- Korzeniewski K. Prevalence of intestinal parasitic infections in the population of Central Asia on the example of inhabitants of Eastern Afghanistan. Przegl Epidemiol. 2016; 70:563–573. PMID: 28221011
- Korzeniewski K, Smolen A, Augustynowicz A, Lass A. Diagnostics of intestinal parasites in light microscopy among the population of children in eastern Afghanistan. Ann Agric Environ Med. 2016; 23:666– 670. https://doi.org/10.5604/12321966.1226864 PMID: 28030941
- 44. Abrar UI Haq K, Gul NA, Hammad HM, Bibi Y, Bibi A, Mohsan J. Prevalence of *Giardia intestinalis* and *Hymenolepsis nana* in Afghan refugee population of Mianwali district, Pakistan. Afr Health Sci. 2015; 15:394–400. https://doi.org/10.4314/ahs.v15i2.12 PMID: 26124784
- 45. Duda A, Kosik-Bogacka D, Lanocha-Arendarczyk N, Kolodziejczyk L, Lanocha A. The prevalence of Blastocystis hominis and other protozoan parasites in soldiers returning from peacekeeping missions. Am J Trop Med Hyg. 2015; 92:805–806. https://doi.org/10.4269/ajtmh.14-0344 PMID: 25732683
- US Department of Health and Human Services, CDC. Domestic Intestinal Parasite Guidelines. [cited 2019 Feb 10]. Available from: https://www.cdc.gov/immigrantrefugeehealth/guidelines/domestic/ intestinal-parasites-domestic.html.
- 47. US Department of Health and Human Services, CDC. Screening for Sexually Transmitted Diseases during the Domestic Medical Examination for Newly Arrived Refugees. [cited 2020 Jan 8]. Available from: https://www.cdc.gov/immigrantrefugeehealth/guidelines/domestic/sexually-transmitted-diseases/ index.html
- US Department of Health and Human Services, CDC. Infant lead poisoning associated with use of tiro, an eye cosmetic from Nigeria—Boston, Massachusetts, 2011. MMWR Morb Mortal Wkly Rep. 2012; 61:574–576. PMID: 22854626
- 49. US Department of Health and Human Services, CDC. Childhood lead exposure associated with the use of kajal, an eye cosmetic from Afghanistan—Albuquerque, New Mexico, 2013. MMWR Morb Mortal Wkly Rep. 2013; 62:917–919. PMID: 24257200
- McMichael JR, Stoff BK. Surma eye cosmetic in Afghanistan: a potential source of lead toxicity in children. Eur J Pediatr. 2018; 177:265–268. https://doi.org/10.1007/s00431-017-3056-z PMID: 29224187
- Pezzi C, Lee D, Kennedy L, Aguirre J, Titus M, Ford R, et al. Blood Lead Levels Among Resettled Refugee Children in Select US States, 2010–2014. Pediatrics. 2019; 143(5). Epub 2019/04/19. https://doi.org/10.1542/peds.2018-2591 PMID: 30996119.